Author's response to reviews

Title: The prediction of discharge from in-patient psychiatric rehabilitation: a case-control study.

Authors:

Joanna Bredski (jbredski@nhs.net)
Andrew Watson (awatson3@nhs.net)
Debbie A Mountain (debbie.mountain@nhslothian.scot.nhs.uk)
Fiona Clunie (fiona.clunie@nhslothian.scot.nhs.uk)
Stephen M Lawrie (s.lawrie@ed.ac.uk)

Version: 4 Date: 8 August 2011

Author's response to reviews:

Covering letter

Dear Editors,

Re: The prediction of discharge from in-patient psychiatric rehabilitation: a case control-study.

In response to the editorial requests I have copy edited the paper and I hope that the paper now conforms well with the journal style.

Thank you very much again for all of your help so far with this paper.

Yours faithfully,

Dr Joanna Bredski